Reply Reviewer:

The authors presented an interesting manuscript. I fully agree with the authors that the vasculature is an important pathway for tumor cell spread over long distances, and vascular suppositories are a manifestation of tumor invasion of lymphatics or blood vessels. The authors presented a vascular cancer thrombus as a marker characterizing the malignant degree of gastric cancer. This marker has prognostic value and can be used to predict the dynamics of development, as well as the results of treatment of gastric cancer. This study analyzed the results of clinical and pathological tests of 621 patients with gastric cancer. Vascular cancer embolism may be an additional prognostic marker of progression along with the depth of tumor invasion and lymph node metastasis. The manuscript is well illustrated and meets all the requirements of evidence-based medicine.

Thank you very much for the reviewer’s recognition of the quality and level of our manuscript, and thank you for your suggestions and evaluation of our manuscript. We will cooperate with the editor to further improve our manuscript. Thank you for your review work and patience.